In this work, we apply jet transport techniques as implemented in the open-source software package NEOs.jl \citep{Perez2021,Perez-Hernandez2024} to the problem of orbit determination and impact probability predictions for asteroid 2024 PDC25, the hypothetical scenario for the 9th Planetary Defense Conference 2025. From the astrometry file available for 2024 PDC25 at Epoch 1 \citep{2024pdc25-astrometry}, we perform a preliminary orbit determination using only observations from the first three nights available in the file. We then perform an orbit determination to the full astrometry dataset until Epoch 1, taking as initial guess the output from the preliminary orbit determination step. We use the latter orbital fit to propagate the asteroid region of uncertainty until its close approach to the Earth in 2041. Finally, we provide impact probability figures for the close approach to the Earth in 2041, and assess how observations in August 2025 will affect these predictions.

% This is the \LaTeX~abstract preparation template for the IAA Planetary Defense Conference. Each abstract will be evaluated by the members of the program committee and scored based on quality, originality, and relevance to conference. The abstracts will be ranked according to a score and selected for presentation based on the available space. Please note that failure to follow this format will lead to rejections of the paper without review. If selected for presentation a paper or an extended abstracts will be is required.

% The abstract length should be between 250 and 500 words. It must contain the paper title, the author names, their corresponding affiliation, postal and e-mail addresses. The abstract shall be prepared using the following format:

% \begin{itemize}
% 	\item Paper: standard A4 paper (21.0 $\times$ 29.7 cm)
% 	\item Margins: 2.54 cm (all sides)
% 	\item Title: 12 pt bold, Arial, centered, all capital letters
% 	\item Author names: 12 pt bold, Arial, centered
% 	\item Affiliation: 12 pt italic, Arial, centered
% 	\item Abstract body: 12 pt, Arial, justified with single line spacing
% 	\item Equations, figures, literature may be included with references in the body text
% \end{itemize}

% Abstracts must be written in English and should be submitted to the IAA as Microsoft Word files (.doc, .docx) or Portable Document Format files (.pdf) within the deadline. All papers must be submitted online to iaapdc (at) iaamail.org  It is recommended to proof read the paper before submission. Word format files are preferred but pdf format files will also be accepted.

% Notification about the review process is planned according calendar. This will give the possibility to profit from the “early registration fee” of the conference. Please note that the presenting author of each accepted abstract has to register to the conference.

% Example reference citation~\cite{SampleRef3}. Example figure reference~\ref{fig:fig1}. Example table reference~\ref{tab:tab1}.

% \begin{table}[h!]
%     \centering
%     \caption{\small{Caption goes here.}}
%     \label{tab:tab1}
%     \begin{tabular*}{0.99\columnwidth}{p{1.12in} p{1.12in} p{1.12in}}
%         \toprule
%         Column Hdg & Column Hdg & Column hdg\\
%         \midrule
%         Data & Data & Data \\
%         Data & Data & Data \\
%         Data & Data & Data \\
%         Data & Data & Data \\
%         \bottomrule
%     \end{tabular*}
%     %\vspace{0.02in} %adjust spacing after table before text
% \end{table}

% \begin{figure}[h!]
%     \begin{center}
%         \includegraphics[width=0.5\textwidth, height=3.65in]{placeholder_fig}
%         \caption{\small{Caption goes here.}}
%         \label{fig:fig1}
%     \end{center}
% \end{figure}


\textbf{Comments:}

\textit{We prefer a oral presentation.}

\endinput

